Author's response to reviews

Title: Adherence to treatment in children and adolescents with cystic fibrosis: a cross-sectional, multi-method study investigating the influence of beliefs about treatment and parental depressive symptoms

Authors:

Nicola A Goodfellow (Nicola.Goodfellow@nictu.hscni.net)
Ahmed F Hawwa (a.hawwa@aston.ac.uk)
Alastair JM Reid (alastair.reid@belfasttrust.hscni.net)
Rob Horne (r.horne@ucl.ac.uk)
Michael D Shields (m.shields@qub.ac.uk)
James C McElnay (j.mcelnay@qub.ac.uk)

Version: 5
Date: 2 April 2015

Author's response to reviews: see over
Dear Editor,

RE: Research article 8305800721443991

The formatting of the manuscript has been revised as per editorial request.

We hope that with these amendments that the paper is now suitable for publication.

Yours sincerely,

Professor J.C. McElnay
Professor of Pharmacy Practice